\documentclass{article}

\setlength{\parindent}{0pt}
\setlength{\voffset}{-0.5in}
\setlength{\textheight}{9in}

\pagestyle{empty}

\begin{document}

\textbf{MATH 425: Complex Variables \hfill Spring 2002}

\bigskip
\centerline{\textbf{Course Information}}

\bigskip
\textsc{Instructor:} Dennis Snow, \texttt{snow.1@nd.edu}

\bigskip
\textsc{Office Hours:} TH 9:30--11:00, HAYE 142, and by appointment

\bigskip
\textsc{Text:} Fisher, \emph{Complex Variables}, 2nd Edition, Chapters 1--3.

\bigskip
\textsc{Class Time:} MWF 3:00--3:50 in HAYE 125.

\bigskip
\textsc{Exams:} There will two take-home exams and a final
on the following dates:

\begin{center}\begin{tabular}{ll}
Exam I  &Friday, Feb 22, due Mon Feb 25 \\
Exam II &Friday, Apr 12, due Mon Apr 15 \\
Final   &Monday, May 6, 4:15--6:15 P.M. \\
\end{tabular}\end{center}

\bigskip
\textsc{Quizes:} There will be a short quiz every Friday, except Jan~18,
Feb~22, Mar~29, and Apr~12.

\bigskip
\textsc{Honor Code:} The Honor Code is in effect for all exams and
quizzes.

\bigskip
\textsc{Grades:} Grades will be based on a total of 450 points broken
down as follows:

\begin{center}\begin{tabular}{lclr}
Quizes      & 10 @ 10  &=& 100\\
Exams I, II &  2 @ 100 &=& 200\\
Final       &          &=& 150\\
\end{tabular}\end{center}

\bigskip

\centerline{\textbf{Suggested Exercises}}

\bigskip

1.1 Complex numbers, p.9:1--19odd

1.2 Some geometry, p.20:1--29odd

1.3 Subsets of the plane, p.28:1--17odd

1.4 Functions and limits, p.41:1--23odd,31--41odd

1.5 Exponential, log, and trigonometric functions, p.53:1--29odd

1.6 Line integrals and Green's Theorem, p.73:1--11odd,15

2.1 Analytic functions: CR equations, p.85:1--11odd,15,17,21

2.2 Power series, p.103:1--23odd

2.3 Cauchy's Theorem and Formula, p.116:1--15odd

2.4 Consequences of Cauchy's Theorem, p.133:1--23odd

2.5 Isolated singularities, p.150:1--15odd

2.6 Residue Theorem and integrals, p.167:1-23odd

3.1 Zeros of an analytic function, p.179:1--15odd

3.2 Maximum modulus and mean-value, p.194:1--11odd

3.3 Linear fractional transformations, p.204:1--11odd

[3.4 Conformal mapping, p.218:1,2,7,10,15]

[3.5 Riemann mapping theorem, p.241:1--11odd]

\end{document}
